Author’s response to reviews

Title: Prevalence of diarrhoea and associated risk factors among children under five years old in Pader District, Northern Uganda

Authors:

Stephen Omona (omonastephen@gmail.com)
Geoffrey Malinga (malingageoffrey@yahoo.com)
Robert Opoke (opokerobert@gmail.com)
Geoffrey Openy (gopeny@gmail.com)
Robert Opiro (robopiro@gu.ac.ug)

Version: 2 Date: 31 Dec 2019

Author’s response to reviews:

20th December 2019

Dear Zhongjie Shi,

Re: Response to review comments on our resubmitted Manuscript INFD-D-19-01866R1

Thank you once again for considering our revised manuscript potentially acceptable for publication in BMC Infectious Diseases. Please find below our point-by-point specific responses to the review comments. We sincerely hope that the revisions we have made makes this manuscript worthy for publication in BMC Infectious Diseases.

Editor Comments:

1. Please state whether the maps depicted in figure 1 are your own or taken from another source.
Response: We have now indicated that the map was produced by us the authors (See LN 383-384 Page 15).
2. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Response: The manuscript has now been resubmitted as a clean copy without any tracked, colored or highlighted text.

Sincerely,

Robert Opiro, PhD
Corresponding author
E-mail: robopiro@gu.ac.ug